Author’s response to reviews

Title: Feasibility of designing, manufacturing and delivering 3D printed ankle-foot orthoses: a systematic review

Authors:

Elizabeth Wojciechowski
(elizabeth.wojciechowski@health.nsw.gov.au; elizabeth.wojciechowski@sydney.edu.au)

Angela Chang (acha6045@uni.sydney.edu.au)

Daniel Balassone (daniel.balassone@health.nsw.gov.au)

Jacqueline Ford (jacqueline.ford@health.nsw.gov.au)

Tegan Cheng (tegan.cheng@sydney.edu.au)

David Little (david.little@health.nsw.gov.au)

Manoj Menezes (manoj.menezes@health.nsw.gov.au)

Sean Hogan (sean.hogan@health.nsw.gov.au)

Joshua Burns (joshua.burns@sydney.edu.au)

Version: 2 Date: 29 Jan 2019

Author’s response to reviews:

Editors comments

i) Please provide some references to content in the final paragraph of the 'Background'
Amended (page 4, line 73 and line 79 and page 5 line 81)

ii) Remove commas after 'Faustini and colleagues [12],' (line 281) and 'Schrank and colleagues [15],' (line 286).
Amended (page 13, line 281 and and page 14, line 286)

iii) Figure 1, please make the box slightly bigger under the 'eligibility' section as it is cutting off the bottom of the words/numbers (where it says 'design (n=2)').
iv) Please check spacing before and after sentences. It appears that there are some single and double spaces being used inconsistently (this may not be the case but please double check).

Checked